Author's response to reviews

Title: Diffuse large B cell lymphoma derived from nodular lymphocyte predominant Hodgkin lymphoma presents with variable histopathology

Authors:

Sylvia Hartmann (s.hartmann@em.uni-frankfurt.de)
Mine Eray (mine.eray@uta.fi)
Claudia Döring (c.doering@em.uni-frankfurt.de)
Tuula Lehtinen (tuula.lehtinen@pshp.fi)
Uta Brunnberg (uta.brunnberg@kgu.de)
Paula Kujala (paula.kujala@fimlab.fi)
Martine Vornanen (martine.vornanen@fimlab.fi)
Martin-Leo Hansmann (martin.l.hansmann@em.uni-frankfurt.de)

Version: 3
Date: 2 May 2014

Author's response to reviews:

Outline of specific responses to the comments:

EDITORIAL REQUEST:

1). Please could you anonymise figure 4, please remove the age and gender of the patient in your description of the figure. Please anonymise the four specific cases that you describe at the bottom of page 9 (i.e. please remove ages).

Age and gender of the patient has been deleted in favour of a more generalized description of figure 4. On page 9, the age of the patients has been replaced by the median age of the patients in order to anonymise these.

2). Please could you provide emails for all authors on your title page.

Emails of all authors are provided on the title page.

3). Please provide supplementary figure legends after your current figure legends section.

A supplementary figure legend for supplementary figure 1 has been included after the figure legend section.